Author's response to reviews

**Title:** Reducing salt intake for prevention of cardiovascular diseases in high-risk patients by advanced health education intervention (RESIP-CVD study), northern Thailand: study protocol for a cluster randomized trial

**Authors:**

- Myo N Aung (dr.myonyeinaung@gmail.com)
- Motoyuki Yuasa (moyuasa@juntendo.ac.jp)
- Saiyud Moolphate (ritbkk@bkk7.loxinfo.co.th)
- Supalert Nedsuwan (supalert@gmail.com)
- Hidehiro Yokokawa (yokokawa@pa3.so-net.ne.jp)
- Tsutomu Kitajima (kitajima@ks.kyorin-u.ac.jp)
- Kazuo Minematsu (kminema@juntendo.ac.jp)
- Susumu Tanimura (tanimura@hyo-med.ac.jp)
- Hiroshi Fukuda (hiro----@kt.rim.or.jp)
- Yoshimune Hiratsuka (yoshi-h@tkf.att.ne.jp)
- Koichi Ono (kono@juntendo.ac.jp)
- Sachio Kawai (chyabo@juntendo.ac.jp)
- Eiji Marui (marui@juntendo.ac.jp)

**Version:** 5 **Date:** 1 August 2012

**Author's response to reviews:**

Dear Sir,

I have amended the title to conform to the Trials style

Reducing salt intake for prevention of cardiovascular diseases in high-risk patients by advanced health education intervention (RESIP-CVD study), northern Thailand: study protocol for a cluster randomized trial